Author’s response to reviews

Title: Influence of weight gain, according to Institute of Medicine 2009 recommendation, on spontaneous preterm delivery in twin pregnancies

Authors:

Paola Algeri (p.algeri@campus.unimib.it)
Francesca Pelizzoni (f.pelizzoni@virgilio.it)
Davide Bernasconi (davide_bernasconi@libero.it)
Francesca Russo (f.russo84@gmail.com)
Maddalena Incerti (maddalena.incerti@gmail.com)
Sabrina Cozzolino (cozzolinosabrina@libero.it)
Salvatore Andrea Mastrolia (mastroliasa@gmail.com)
Patrizia Vergani (patrizia.vergani@unimib.it)

Version: 4 Date: 13 Dec 2017

Author’s response to reviews:

To

Editor-in-Chief

BMC Pregnancy and Childbirth

RE: Ms. No. PRCH-D-16-00410R3 - Influence of weight gain, according to Institute of Medicine 2009 recommendation, on spontaneous preterm labor in twin pregnancies

We thank the Editor for his prompt evaluation of the last revised version of our manuscript and feel very happy and proud about the fact that it has been considered as acceptable for publication in BMC Pregnancy and Childbirth. In addition, we would like to thank the Editorial board for positively considering our availability in answering all Editorial and Reviewers' comments in all rounds of revision.

Listed below is a point-by-point response to the technical comments provided by the Editor, and attached are the files of the revised version of the manuscript.
Best regards,

Paola Algeri, MD
Department of Obstetrics and Gynecology
Fondazione MBBM, San Gerardo Hospital
University of Milano Bicocca
Via Pergolesi, 33
20900, Monza, Italy
Email p.algeri@campus.unimib.it

Technical Comments

Comment #1: Authors' Contributions section: Please indicate authors by their initials rather than full name, e.g. JD for John Doe.

Response to comment #1: Authors initials were provided in place of full names within the Author's Contributions section

Comment #2: Also, please include a statement at the end of this section that "All authors have read and approved the final version of the manuscript"

Response to comment #2: We thank the Editor for his suggestion, and added the requested statement at the end of the Author's Contributions section

Comment #3: At this stage, please upload a final, clean version of the manuscript, with no highlights or tracked changes

Response to comment #3: A clean version of the manuscript, with no highlights or tracked changes has been uploaded within the submission system, as per editorial request